Author’s response to reviews

Title: Study of laboratory staff’s knowledge of biobanking in Côte d'Ivoire

Authors:
Ambroise Kouamé Kintossou (kkintossou@gmail.com)
Mathias Kouamé N'Dri (ndri_kmat@yahoo.fr)
Marcelle Money (ettienmarcelle@yahoo.fr)
Souleymane Cisse (souleymane_6c@yahoo.fr)
Simini Doumbia (doumbiasim@yahoo.fr)
Soumahoro Man-Koumba (mksoumahoro@pasteur.ci)
Amadou Founzégué Coulibaly (founzegue@yahoo.fr)
Joseph Allico Djaman (josephdjaman@gmail.com)
Mireille Dosso (mireilledosso@yahoo.fr)

Version: 2 Date: 01 Sep 2020

Author’s response to reviews:

1. The entire document has been revised in English as indicated.
2. The survey form was added to the submitted documents as indicated.